Author's response to reviews

Title: Breast cancer screening: evidence of benefit depends on the method used

Authors:

Philippe Autier (Philippe.autier@i-pri.org)
Mathieu Boniol (mathieu.boniol@i-pri.org)

Version: 12 Date: 23 October 2012

Author's response to reviews:

Lyon, 23th of October 2012

MS: 1812492806841872

Opinion
Breast cancer screening: evidence of benefit depends on the method used
Philippe Autier and Mathieu Boniol
BMC Medicine

Dear Ms Barnard,

Please receive the re-revised version of our opinion paper with outstanding corrections. We added a sentence at the end of the section on case-control studies and a reference in the section on breast cancer mortality.

We also added "Both authors approved the final version of the manuscript" in the author's contribution.

We thank you for the opportunity of having our opinion considered by BMC Medicine.

With best regards,

Philippe Autier, MD